Author's response to reviews

Title: An exploratory spatial analysis of geographical inequalities of birth intervals among young women in the Democratic Republic of Congo (DRC): a cross-sectional study

Authors:

   Tobias F. Chirwa (tobias.chirwa@wits.ac.za)
   Felly K Kinziunga (fellyki@gmail.com)
   Jocelyn N Mantempa (jocelynmantempa@gmail.com)
   Joseph D Kandala (josephdesirekandala@yahoo.fr)
   Ngianga-Bakwin Kandala (N-B.Kandala@warwick.ac.uk)

Version: 3
Date: 14 June 2014

Author's response to reviews: see over
Dear Editor,

Thank you for the editor’s and reviewer’s comments which we believe have improved and strengthened the quality of the manuscript. On behalf of the co-authors, I sincerely appreciate this feedback. Thank you for giving us the opportunity to revise and resubmit the manuscript.

We would like to resubmit the revised manuscript entitled “An exploratory spatial analysis of geographical inequalities of birth intervals among young women in the Democratic Republic of Congo (DRC): a cross-sectional study”.

We have examined each and every comment from the editor and reviewer and the details of our responses to their comments, suggestions and questions have been submitted under the response section, and taken into account in the updated manuscript. We hope we have satisfactorily addressed the questions and suggestions pointed out.

Yours sincerely,

Tobias F. Chirwa, PhD

Corresponding Author